Author's response to reviews

Title: A migraine variant with abdominal colic and Alice in wonderland syndrome: a case report and review

Authors:

sherifa A Hamed (hamed_sherifa@yahoo.com)

Version: 5 Date: 18 December 2009

Author's response to reviews: see over
The manuscript (MS: 2377696722882079) entitled “A Migraine Variant with Abdominal Colic and Alice in Wonderland Syndrome: A Case Report and Review” was revised taking into consideration the formatting changes requested as follow:

- The abstract has been put into a correct structure.
- The abstract is now identical in the manuscript and submission system.
- I amended the title as requested.
- Running title was removed from page 1.
- Keywords were removed.
- All highlighting/tracking and colored text were removed.
- Structure of the manuscript was done as requested: abstract, case presentation, discussion, conclusions, competing interests, acknowledgment, references, tables.
- The vertical visible lines were removed from the tables.
- The whole text was revised for typographical and grammar errors.
- Please keep the title discussion in the text as its removal will significantly impair the appeal of the manuscript as a whole.

Sincerely yours,
Sherifa
Sherifa Ahmed Hamed, MBBch., MSc., M.D.
Consultant Neurologist
Associate Professor, Department of Neurology and Psychiatry, Assiut University Hospital, Assiut, Egypt
P.O.Box 71516
Telephone: +2 088 2371820
Fax : +2 088 2333327
+2 088 2332278
email: hamed_sherifa@yahoo.com